Author’s response to reviews

Title: PREVALENCE OF ANEMIA AMONG ADULTS AT HAWASSA UNIVERSITY REFERRAL HOSPITAL, SOUTHERN ETHIOPIA.

Authors:

Misganaw Mengesha (miganawbs@gmail.com)
Gezahegn Dadi (bekeledadi@gmail.com)

Version: 4 Date: 16 Nov 2018

Author’s response to reviews:

REVISION 3

Thank you reviewer 1 and 2 for accepting the work.

Reviewer 3

Controls: This is a simple prevalence study on anemic patients, which the study didn’t have controls on healthy subjects data. Therefore, initially it was not the objective of the study to include controls and do anemia findings from healthy individuals. Statistics: In my view appropriate statistical methods and analysis were carried out, while the reviewer didn’t implicate where additional statistics were required in specific.

Editor: Track changes and highlights removed. Format was maintained according to the journal style and comments from editor.